Author’s response to reviews

Title: Higher cerebral oxygen saturation may provide higher urinary output during continuous regional cerebral perfusion

Authors:

Takashi Miyamoto (yonomiyataka@msn.com)
Kagami Miyaji (kagami111@aol.com)
Hirotsugu Okamoto (okasuke@med.kitasato-u.ac.jp)
Satoshi Kohira (me-ksat@kitasato-u.ac.jp)
Takahiro Tomoyasu (kingbon0716@yahoo.co.jp)
Nobuyuki Inoue (nobuyuki-i@mail.goo.ne.jp)
Kuniyoshi Ohara (oharak@med.kitasato-u.ac.jp)

Version: 7 Date: 29 October 2008

Author’s response to reviews:

Thank you very much for reviewing our manuscript number: MS: 5216423921949676.

We extensively revised it according to major and minor changes of the reviewers.

We hope to have met your requirements that greatly improved this manuscript.

Yours gratefully and faithfully,

Takashi Miyamoto, PhD, MD.